Author's response to reviews

Title: A case of VACTERL spectrum presenting with Portal Hypertension: a case report

Authors:

Dilli Raj Bhurtel (dilibhurtel@yahoo.com)
Ignatius Losa (ignatius.losa@echeshire-tr.nwest.nhs.uk)

Version: 6 Date: 12 February 2010

Author's response to reviews:

Dear Editorial team,

I have revised the recently submitted manuscript. I have made following changes:

1. Authors name: Middle name of the first author was missing, second author's first name was spelled incorrectly I have corrected them.

2. In terms of the differences in the abstract in the manuscript and the submission system: Its been updated on the online submission system so it matches with the manuscript.

3. The references numbers are put on square brackets as suggested.

4. The date and the numbers are removed from the images.

Many thanks for your expected help,

Dilli Raj Bhurtel